Author's response to reviews

Title: Home visits and Village Health Team consultations, a well-accepted method for improving communication about maternal and newborn care practices in Masindi and Kiryandongo districts, Uganda

Authors:

Richard Mangwi Ayiasi (mangwirichard@gmail.com)  
Lynn Muhimbura Atuyambe (atuyambe@musph.ac.ug)  
Juliet Kiguli (jkiguli@musph.ac.ug)  
Christopher Orach Garimo (cgorach@musph.ac.ug)  
Patrick Kolsteren (pkolsteren@itg.be)  
Bart Criel (bcriel@itg.be)

Version: 2  
Date: 4 March 2015

Author's response to reviews:

The revisions we have made are related to formatting of the manuscript as required by the journal: numbering all lines in the manuscript; including a section on keywords; authors details; and abbreviations.